Author’s response to reviews

Title: Diagnostic value of medical thoracoscopy in malignant pleural effusion

Authors:

Yan-Bing Wu (wuyanbing7755@163.com)
Li-Li Xu (lily_5809@qq.com)
Xiao-Juan Wang (xjwang730715@hotmail.com)
Zhen Wang (wz_0316@126.com)
Jun Zhang (zhangjun19972005@163.com)
Zhao-Hui Tong (tongzhaohuicy@sina.com)
Huan-Zhong Shi (shihuanzhong@sina.com)

Version: 1 Date: 05 Nov 2016

Author’s response to reviews:

1. The textual overlap with our previously published works has been corrected in this re-submission.

2. Sorry for that mistake, the ethical approval was approved, what we mean in the text is "information consents were not sought since this is a retrospective study". We have correct it in this new manuscript.